

Received November 9, 2020, accepted November 22, 2020, date of publication November 30, 2020, date of current version December 11, 2020.

Digital Object Identifier 10.1109/ACCESS.2020.3041247

Electric Power Grids Under High-Absenteeism Pandemics: History, Context, Response, and Opportunities

BENJAMIN WORMUTH^{®1}, (Student Member, IEEE), SHIYUAN WANG^{®1}, (Member, IEEE), PAYMAN DEHGHANIAN^{®1}, (Senior Member, IEEE), MASOUD BARATI^{®2}, (Senior Member, IEEE), ABOUZAR ESTEBSARI^{®3}, (Member, IEEE), TIAGO PASCOAL FILOMENA⁴, MOHAMMAD HEIDARI KAPOURCHALI^{®5}, (Member, IEEE), AND MIGUEL A. LEJEUNE^{®1,6}

¹Department of Electrical and Computer Engineering, The George Washington University, Washington, DC 20052, USA

²Department of Electrical and Computer Engineering, University of Pittsburgh, Pittsburgh, PA 15261, USA

⁶Department of Decision Sciences, School of Business, The George Washington University, Washington, DC 20052, USA

Corresponding author: Payman Dehghanian (payman@gwu.edu)

This work was supported in part by the U.S. National Science Foundation (NSF) under Grant ICER-2022505 and Grant CNS-1951847 (for Payman Dehghanian) and ICER-2022705 (for Mohammad Heidari-Kapourchali). The work of Masoud Barati was supported in part by NSF under Grant ECCS-1711921. The work of Miguel A. Lejeune was supported by the U.S. Office of Naval Research under Grant N000141712420.

ABSTRACT Widespread outbreaks of infectious disease, i.e., the so-called pandemics that may travel quickly and silently beyond boundaries, can significantly upsurge the morbidity and mortality over large-scale geographical areas. They commonly result in enormous economic losses, political disruptions, social unrest, and quickly evolve to a national security concern. Societies have been shaped by pandemics and outbreaks for as long as we have had societies. While differing in nature and in realizations, they all place the normal life of modern societies on hold. Common interruptions include job loss, infrastructure failure, and political ramifications. The electric power systems, upon which our modern society relies, is driving a myriad of interdependent services, such as water systems, communication networks, transportation systems, health services, etc. With the sudden shifts in electric power generation and demand portfolios and the need to sustain quality electricity supply to end customers (particularly mission-critical services) during pandemics, safeguarding the nation's electric power grid in the face of such rapidly evolving outbreaks is among the top priorities. This paper explores the various mechanisms through which the electric power grids around the globe are influenced by pandemics in general and COVID-19 in particular, shares the lessons learned and best practices taken in different sectors of the electric industry in responding to the dramatic shifts enforced by such threats, and provides visions for a pandemic-resilient electric grid of the future.

INDEX TERMS Absenteeism, COVID-19, electric power grid, lock-down, pandemic, resilience.

I. INTRODUCTION

Electric power grids, as the most complex man-made cyberphysical system to date, have been traditionally designed to operate reliably under normal operating conditions and withstand potential outage-inducing events. However, power grids are constantly confronting a wide range of unpredictable and highly-uncertain threats that could impede preservation of sustainable, reliable, and high-quality electricity needed to electrify modern society. Future societal growth and inno-

The associate editor coordinating the review of this manuscript and approving it for publication was Ning Kang^(b).

vations are now hindered by (i) geographically-vast area of coverage translated into an expanded exposure of the electricity grid to a wide range of threats, (ii) the rushing arrival of uncertain renewables and the intensified stochasticity in the electric grid generation portfolios, and (iii) the elevated number and severity of the extreme natural disasters (e.g., storms, hurricanes, earthquakes, wildfires, etc.) and manmade cyber-attacks. Safeguarding the nation's electric power grid and ensuring a continuous, reliable, and affordable supply of energy in the face of such extremes, commonly known as high-impact low-probability (HILP) events, are among the top priorities for the electric power industry. While some of

³School of the Built Environment and Architecture, London South Bank University, London SE1 0AA, U.K.

⁴Operations Research and Finance, Federal University of Rio Grande do Sul, Porto Alegre, Brazil

⁵Department of Electrical Engineering, University of Alaska Anchorage, Anchorage, AK 99508, USA



FIGURE 1. Global pandemics and deaths in the 20th and 21st centuries [1], [2].

these extremes are difficult to predict, the electric industry generally recognizes the need to build in system survivability to keep the lights on at all times and to develop analytical tools for online monitoring, detection, verification, and mitigation of such threats and many other emergencies that adversely affect the power grid operation and control. Resilience of the electricity grid is now more critical than ever to people's well-being and every aspect of our increasingly-electrified economy.

The frequency of pandemics has been on the rise over the past 20 years (see Figure 1) [3]. Examples include the 2002 Severe Acute Respiratory Syndrome (SARS), 2009 Novel Influenza A Virus Subtype H1N1, 2003-2016 Ebola Virus Disease (EVD), and the ongoing Coronavirus Disease (COVID-19), with a large rate of fatality and occupation around the world. The emergence of pandemics is primarily driven by advancements in urbanization, intensified global travel and integration, drastic shifts in land use, and greater exploitation of the natural environment. While significant efforts and vigorous measures have been taken to identify and limit the emerging outbreaks that might lead to pandemics, various sectors of the modern societies and critical infrastructure around the globe are still vulnerable to and ill-prepared for these extremes. Pandemics have been observed to result in abrupt, co-evolved, and sometimes challenging shifts in socio-ecological ecosystems, businesses, and built environments [4].

Electric power grids are not an exception. The planning, operation and control of the grid is impacted by such "high absenteeism" events in different sectors and on multiple scales. From health emergencies that could rapidly and severely limit the available workforce in electric industry to supply-chain disruptions leading to postponed installations and delayed maintenance implementations, the electric industry has to take all necessary actions to (i) keep the lights on throughput the pandemic emergency and (ii) ensure the health and well-being of the available personnel and on-field workforce. Different from the business-as-usual practices, drastic changes and sudden shifts in the use of electricity by the industrial, commercial, and residential sectors during pandemics accrue the complexity of taking decisions under such highly volatile contexts.

This paper is a wide-ranging overview of the effects of pandemics on electric utilities, power grid, and load demand. By covering the effects of historical pandemics all the way through a plan in response to future pandemics, this paper provides context for the COVID-19 pandemic's impacts on the utility industry and the demand for electricity. Drawing on all of these experiences, the paper aims to share the lessons learned and best practices in the fight between the power engineering and pandemics, shedding lights on what it takes to realize the pandemic-resilient power grids of the future.

The remainder of the paper is structured as follows. Section II provides background information on several major pandemics in the past 20 years and discusses how electric power grids responded to the resulting states of emergency. Focusing on the ongoing COVID-19 emergency, Section III presents a global perspective on how electric power grids of different sizes, technological complexity, and geographical classifications around the world have been influenced in 6 months of COVID-19 emergence and evolution. Section IV is devoted to the United States electric power grid and describes (i) the existing vulnerabilities in facing the outbreak, (ii) the actions and policies planned and/or taken by different authorities in the pre-pandemic alert phase and through-pandemic outbreak phase, and (iii) the post-pandemic plans in returning to the business-as-normal phase of operation. Section V introduces the observations on how the COVID-19 pandemic outbreak could impact the interdependent critical infrastructure and lifeline networks (e.g., communication networks, water systems, transportation networks, health sector, etc.) which are driven by the electric power grid and their response to COVID-19 pandemic could shift the electricity consumption patterns across the country. Section VI discusses the challenges and opportunities during and following the COVID-19 pandemic outbreak along with the visions for a pandemicprepared and pandemic-resilient power grids. Section VIII provides the concluding remarks.



FIGURE 2. The load softened and shifted slightly later in the day due the COVID-19 pandemic [13].

II. HISTORICAL PANDEMICS AND ELECTRIC GRIDS

COVID-19 is certainly not the first pandemic to pose challenges to the electric power grid. In the past 20 years, there have been a handful of outbreaks [3] that had the potential to disrupt society and damage the power grid operation (see Figure 1). The main stress on the utility system is absenteeism from workers getting sick [5]. Loss of institutional knowledge is a devastating risk during pandemics. Should enough workers die or become unavailable, the electric utility would have to spend massive sums of money to regain that knowledge. Until COVID-19, none of these past outbreaks had a widespread impact on the power grid operations and field practices. At each pandemic, electric utilities became more prepared to combat such events and revisit their contingency plans.

A. SARS

In November 2002, the Severe Acute Respiratory Virus (SARS) was first discovered in the Guangdong Province in China. In February 2003, the virus resurfaced in Hanoi, Vietnam. In total, there were about 8,000 reported cases and 70 deaths worldwide [6]. With respect to the electric power grid, there was a large scare, but no major implications for reliability of the grid operation. The SARS epidemic has however a positive long-term effect in terms of improving pandemic preparedness. Many electric utilities either created or updated their pandemic preparedness plan in the wake of the crisis. By being proactive and creating a plan before the crisis, the utilities set the stage for a smooth transition through other future health crises.

B. H1N1

After SARS, H1N1 emerged in 2009. It originated from Mexico and spread throughout the globe within a few weeks. Commonly referred to as the Swine Flu, the H1N1 outbreak caused between 151,700 to 575,400 worldwide deaths in the first year [7]. For comparison, as of 28 September 2020, the COVID-19 pandemic has caused 1 million reported deaths worldwide [8]. In 2009, the H1N1 outbreak was not severe enough to stress the power grid, so it did not provide an opportunity to test the infrastructure and pandemic plans for resilience. Apart from the deaths worldwide, the only result of this pandemic was some school closures and public

concern for workers under 65 years old. Unlike other pandemics, 80% of the deaths involved people under 65 years old [7]. The majority of the workforce is under 65 years of age, so this meant special precautions needed to be taken for worker safety. Because of the SARS scare earlier, many electric utilities already had necessary pandemic plans in place. Utility work is often conducted in teams for safety reasons. This continued throughout the H1N1 pandemic for many utility companies. Efforts were made to sequester the field crews from the outside public and other utility workers (such as engineers and managers). One utility even had their field crews add a second barrier to increase the safety radius between the construction work and the public [9]. This helped isolate crews and prevent infection between the public and field crews. Utilities also feared high absenteeism and poor communication methods would hamper their work, but there are no documented instances of this being an issue during the H1N1 pandemic. High absenteeism is an issue because getting work accomplished is very difficult when key stakeholders are out sick at differing times. In severe crises, work may go unfinished if there are not enough healthy workers to complete the projects. Communication platforms have vastly improved since the 2009 H1N1 outbreak, and there are now numerous video chat platforms for communication if workers are working remotely. In many ways, the H1N1 pandemic improved the utility's perspective and response to the COVID-19 pandemic. H1N1 served as a dry run for what was to come, and many electric utilities were able to simply update their H1N1 pandemic response plans for COVID-19.

C. EBOLA

From 2014-2016, 11,310 people died due to an outbreak of the Ebola Virus Disease (EVD) [10]. Primarily concentrated in West Africa, lack of access to reliable electric power made treating EVD particularly challenging. Electricity is crucial to maintain refrigeration, lab work, and ventilators. Without reliable power access, fighting a pandemic is exponentially more challenging. EVD is still a risk in Africa, and fighting it is still a challenge without reliable access to electricity. Many countries in Africa rely on rural micro-grids which lack the reliability found in larger grids.

D. ZIKA

With the Zika outbreak during 2015 and 2016, there was not much impact recorded on utilities and the power grid operation. According to Eric Silagey, Florida Power and Light CEO, the only major change to business as usual practices involved encouraging outdoor workers to wear long sleeves, and increasing the use of bug spray. Because Zika was not usually fatal, it did not have as much impact on electric utility operations [11].

III. COVID-19 AND ELECTRIC POWER GRIDS: A GLOBAL PERSPECTIVE

Across the world, demand for electricity has mostly decreased amid the COVID-19 pandemic. With lock-downs



FIGURE 3. Electricity markets in China during a COVID-19 lock-down period [12].

in place and people working remotely from home, electricity usage was observed lower than usual. Even with many people staying home during the lock-down period, residential power usage only increased by 2.4%. It takes an extraordinary amount of energy to heat, cool, and light the large office buildings found in cities. With high-rise buildings in downtown areas, it takes more energy to pump water and run the elevators. While these energy usages may seem trivial, the aggregated demand for electricity has drastically decreased, and the results have been felt worldwide.

This section discusses the response of the electric industry to COVID-19 across the globe. While the United States has had the most deaths of any country, the virus has affected the electric power grid in other countries and continents in unique ways. This section analyzes such impacts in Canada, Asia, Africa, and the European Union.

A. CANADA

In March 2020, Ontario's Independent Electricity System Operator (IESO) released an updated version of their pandemic plan outlining the response levels for different transmission rates of COVID-19 [14]. The document focuses on the resilient delivery of electricity at all levels, from generation to distribution. Canada had no major disruptions to its electric grid during the COVID-19 outbreak, partly because of the extensive planning. Planning documents have helped utilities keep information well documented and inform employees about all it takes for the lights to be kept on during the COVID-19 emergency [14]. Electric utilities are skilled at planning for emergencies and crises. Many have extensive plans in place for natural disasters, and many have employees holding both a blue-sky job and a storm-duty job. This allows them to mobilize their workforce to restore power as fast as possible. This also instills a culture of flexibility into the employee's work. Employees understand that their sole goal above all else is to keep the lights on. This is a very important culture to maintain during a pandemic as employees work from home and possibly become disconnected from their company. Ontario IESO's document is proof that the Canadian utilities were prepared for the pandemic.

In terms of the load demand, Canada experienced a drastic reduction in power consumption during the strictest lockdown phase of the pandemic. In Ontario, there was a drop of 1000-2000 MW of load demand, or about 10% of their regular demand [17]. This is equivalent to the city of Ottawa no longer using power and imposed serious challenges on the grid. Chiefly, power generation has to be significantly reduced. While this is possible, it places new challenges on the ISO operator during an already stressful period. Generation sources have to be significantly throttled. The grid is highly dynamic within its limits, but such drastic drops significantly stresses the system.

In addition to lower load values, the traditional duck curve for daily power consumption in Ontario, Canada has shifted (see Figure 2). While the traditional curve begins to rise around 7 AM, the curve now rises later in the morning [13], [17]. In addition, the evening peak has flattened as compared to the traditional setting where there would be a peak around 5-6 PM as people get home from work. These changes can challenge the system operators responsible for ensuring a reliable delivery of power to end-use consumers.

B. ASIA

The first confirmed cases of COVID-19 appeared in Wuhan, China in mid-January 2020. Almost immediately, the affected areas locked down. According to the IHS Markit [12], the electricity demand dropped 7.8% for January and February throughout China. Electricity use in industrial facilities and textile industry dropped around 12% and 30%, respectively (Figure 3) [12]. The response was felt primarily in the Hubei province, and the electricity demand remained fairly resilient outside of the originating province. By April, the demand for electricity again ticked upward nationwide [18]. Such large swings in industrial power usage impact the power grid and the utility operations because electric utilities usually charge large industrial customers for both the real and reactive power delivered. Residential customers are only charged for their real power usage, so even if the demand was an equal shift from industrial to residential





FIGURE 4. Road-map for HVDC transmission lines in China's national power grid [15].

TABLE 1.	Current HVDC	Projects in	China [16].
----------	--------------	--------------------	-------------

Project	Capacity	Length (km)	Starting Location	Ending Location	Completion Date (estimated)
Shaanxi-Hubei Line	14 GW	3,000	Shaanxi	Hubei	2021
Yazhong-Nanchang Line	8 GW	1,700	Yazhong	Nanchang	unknown
Shaanbei-Wuhan Line	8 GW	1,100	Shaanbei	Wuhan	unknown
Qinghai-Henan Project	8 GW	1,500	Qinghai	Henan	unknown



FIGURE 5. Expected New annually-added installations: pre-COVID (2019) vs. during-COVID (until second half of 2020) in India [16].

customers, some electric utilities would still be earning less than before.

Following the first pandemic surge, China is poised for a massive economic growth in the renewable sector [19].

COVID-19 has pushed governments further towards unprecedented debt levels, and this will no doubt impact the transition to renewable energy sources. The International Energy Agency (IEA) estimates that the global solar photovoltaic deployment will decline around 8% in 2020 due to the financial crunch the pandemic has caused [18]. At the end of the day, renewable deployment is still expensive for governments, and the structural upgrade of the electric grid is seen as non-essential during a global pandemic. In China, however, the transition to renewable energy sources has continued throughout the pandemic. China manufactures many of the world's solar panels. In addition, solar generation grew 12% year over year [12]. Chinese companies will experience massive growth in the following years in part because countries like India have delayed solar installation. In the coming years, these delays will drive up demand for renewable solutions. China has focused most of its recovery efforts on maximizing Gross Domestic Product. In March 2020, State Grid Corporation of China, China's national power grid, announced its investment of \$26 billion into installing High Voltage Direct Current (HVDC) lines throughout the



FIGURE 6. Summary of support measures provided across Africa for energy sector reinforcement during COVID-19 [20].

country [21] (see Figure 4). Some of the current HVDC projects in China are shown in Table 1. HVDC lines allow for more remote renewable power generation such as wind and solar, since increased HVDC capacity allows for power transfer over longer distances with lower power losses.

In India, the ongoing pandemic has delayed an estimated 3000 MW of solar and wind deployment across the country which were planned to be achieved in 2020 (see Figure 5) [20], [22]. Delaying renewable energy installations is a phenomenon that is seen in numerous countries over the past 6 months of COVID-19 progression. Many renewable projects are partially government-funded, and the government funding for energy projects dried up with the global pandemic. The capital investments for renewable energy implementation was moved to funding hospitals and health care services, taking care of the sick, and supplying personal protective equipment, such as masks and gloves.

C. AFRICA

Sub-Saharan Africa is facing unique challenges with its power grid. According to the World Bank, "only 28% of healthcare facilities have reliable electric power, and only 43% of the population is electrified at all," posing unique challenges for health care officials [23]. Like under Ebola, necessary health measures during COVID-19 are challenging to implement without reliable electric access. That being said, much of Africa has fared comparably well under the COVID-19 pandemic. Many African countries have lower case numbers than the rest of the world, except for some localized hot-spots in Gabon, South Africa, and Libya [24]. Research is ongoing to determine reasons and the impacts.

Another challenge for Africa is the implementation of offgrid solar projects. Some of these projects require outside investment from capital investment firms beyond its continental boundaries. With the COVID-19 being a global pandemic with severe economic implications, there is a chance that those firms will less likely be able to fund projects. Thus far, this has not been the case, but the pandemic may pose long

215732

term challenges if investment capital becomes harder to find. In many African countries, the government has stepped in to help citizens manage their electricity bills, and to help utility companies stay in business during COVID-19 pandemic. Support measures range from support to oil and natural gas to subsidizing residents' energy bills. Figure 6 shows the various support measures throughout the continent. In total, 23 support measures have been implemented.

D. EUROPE

Europe was considered the center of the global COVID-19 outbreak by the World Health Organization (WHO) as of March 13, 2020. In response to the pandemic, different European countries took different approaches. Some countries like Spain, Italy, Belgium and the UK applied severe restrictions, while a few others like the Netherlands and Sweden took less restrictive measures. In the following sub-sections, the impacts of different containment measures on demand for electricity, renewable installations, and electric industry personnel, are discussed.

1) IMPACTS ON DEMAND FOR ELECTRICITY

Comparing the electricity consumption profiles in different European countries reflects the impact of various measures on the households' activities and electricity demand [30]. Figure 7 compares the weekly electricity load consumption in some of the European countries amid COVID-19 measures with a reference week in the year before with a comparable daily average temperature. Except in Sweden with the lowest restrictions in Europe, electricity demand substantially decreased in all other countries after the pandemic was declared and restrictions were implemented. Although lock-downs made the residential load higher than during the normal working days, closure of small businesses with commercial and industrial loads remarkably decreased the total electricity demand. In Belgium, the transmission system operator reported that the loads connected to the power



FIGURE 7. Electricity consumption during the second week of April 2020 compared to a reference week in 2019 for (a) Spain [25], (b) Italy [26], (c) the UK [27], (d) Belgium [28], (e) Netherlands [29], and (f) Sweden [29].

distribution system were reduced more noticeably than those directly connected to the transmission system [31].

Authors in [30] used a Demand Variation Index (DVI) as shown in (1), to compare the average reduction in electricity demand to a reference period.

$$DVI = \frac{\sum_{i=1}^{n} (P_{t_i}^{old} - P_{t_i}^{new})}{n \times \overline{P^{old}}} \times 100$$
(1)

where $P_{l_i}^{new}$ is the load at time t_i , $P_{l_i}^{old}$ is the load at the same time in the reference year, *n* is the number of load samples, and $\overline{P^{old}}$ is the average demand during the reference period. The DVI for the countries under this study including Spain, Italy, Belgium, UK, the Netherlands, and Sweden, were calculated as 25%, 17.7%, 15.6%, 14.2%, 11.6% and -2.1%, respectively. This shows that Netherlands with less restrictions had a low variation compare to the reference period and Sweden with the least restrictions had even a slightly higher electricity demand.

The changes in load patterns are more marked for the daily domestic demand. The reduction in the commercial and industrial demand affects the short-term and mid-term load forecasts, on which electric utilities rely for planning. For example, based on the forecasts, utilities perform reconfiguration in power distribution networks to decrease the network losses, to improve the voltage profile, and to prevent overloading conditions. With such changes in load profiles, the reconfiguration plans need to be reconsidered. Increased losses and inappropriate voltage profiles may be unavoidable in some cases. As the demand for electricity reduces, the share of renewable energy resources increases. However, electric utilities may need to curtail them to avoid overvoltage conditions across the power network.

2) IMPACTS ON RENEWABLE INSTALLATIONS

In Italy, the electricity demand dropped between February 21 and March 21 by more than 18% as the lockdown was imposed on the country [32]. Italy experienced a slower economic recovery than most other countries in the European Union, partly because of the stringent lock-downs making it difficult to return to business as usual. Italy was also one of the main hot spots in Europe, meaning that they had a lot of COVID-19 cases and the virus was centered in the country for a while. In May 2020, the Italian government raised the tax deductible benefit for solar panel installations from 50% to 110% in a bid to encourage solar installations and get the installers back to work. Like most other countries, Italy experienced disruptions in such renewable energy plans [33].

In July 2020, the EU experienced 5% lower demand for electricity than a year prior. According to the IEA, across the EU "lock-down periods have seen new highs in variable renewables contribution as a share of electricity demand [34]." This increase in renewable energy usage is due to multiple reasons, but partly because the overall demand decreased. Lower demand allowed for a higher percentage of renewables to be used in the power generation portfolio.

3) IMPACTS ON ELECTRIC INDUSTRY PERSONNEL

The major impacts of COVID-19 in Europe are felt on the workforce and absenteeism. Power system companies are aware of such effects and established lock-down or teleworking measures for their personnel. In France, just several days after the pandemic outbreak, three Électricité de France S.A employees in nuclear power plants tested positive for coronavirus. Several others who were in contact with those employees were asked to self-isolate for at least two weeks. In Croatia, the main power company advised all employees to perform all activities they can at home. In Belgium, 95% of employees of the transmission system operator Elia started working from home. Some essential employees who had to work from their offices must comply with the enhanced hygiene measures. Power network operators were also halting travels and limiting in-person meetings. The employees of 50Hertz, a German transmission system operator, were no longer allowed to travel to areas classified as risky, and their private travels were also banned. This would substantially impact enhancement and maintenance plans. The control room and other essential personnel were able to operate selfsufficiently [35]. In Greece, the main electricity company named PPC advised its employees not to commute to work by public transportation. They only allowed maximum of 5 customers to enter public offices at the same time. In Slovenia, ELES, the transmission system operator isolated a team of six employees at the national control center for two weeks. They would then be replaced by colleagues who were in home quarantine. Most power companies were following governments' new regulations to protect their employees against COVID-19, and this has resulted in a large percentage of a company's employees working from home. The limitations in work-space, communication challenges, lack of IT skills in senior staff, and possibility of cyber-attacks, is affecting the performance of electric utilities and power companies across the Europe and possibly worldwide.

IV. COVID-19 AND THE U.S. NATIONAL POWER GRID

A. IMPACTS ON ELECTRIC INDUSTRY PERSONNEL

When the COVID-19 pandemic first appeared in the United States, electric utilities were impacted like every other industry. Most non-essential workers were sent home and essential workers masked up to protect themselves at work. This is the same trend seen across almost all other industries in the U.S. The non-essential workers sent home, however, are essential in the long term. Even though their services are not required for the day-to-day operation of the electric grid, their absence will be noticed in the long run. The engineers and managers sent home are responsible for maintenance, testing, and design of modern power grid facilities such as substations, protection schemes, and service. Many of these employees hold certifications in their roles, and such personnel cannot easily be replaced [5]. Recruiting for these positions is also challenging. Upper level engineers are almost exclusively pulled from other electric utilities, but this becomes

.

215734

challenging if every utility is experiencing the same pandemic and shortages. Furthermore, recruiting younger entry-level engineers becomes a challenge. Many of these positions are filled through internship programs or college recruitment fairs in the U.S., and these events were mostly cancelled during the COVID-19 pandemic. Additionally, the loss of critical employees poses challenges in storm response, maintenance, and service planning.

B. IMPACTS ON POWER GENERATION SECTOR

Pandemic planning usually revolves around securing the transmission system against the threats and absenteeism. Little attention is given to securing the fuel supply, even though this is a critical component to power generation [36]. We here discuss aspects of the power generation sector that would be impacted the most during a high-absenteeism pandemic: (i) fossil fuel supply chain and (ii) renewable energy investment, deployment, and operation.

1) PANDEMICS AND THE FOSSIL FUEL

SUPPLY CHAIN IN THE U.S

The fossil fuel chain in the United States is surprisingly fragile, and there is not much coal held in reserve should the supply dry up. Even though there is a large push towards renewable energy resources, many electricity markets near coal mines still burn coal as it is the cheapest fuel available [36]. 40% of the nation's coal production in 2018 came from the low-sulfur mines in Wyoming's Powder River Basin (PRB), which yielded 304.2 million short tons. The total coal production in the Western region amounts to 418.3 short tons, or about 40% of the total US coal production (756.2 short tons) [36], [37]. Most of this coal is hauled by train to distant power plants, some as far away as Georgia. A highabsenteeism interruption in the transportation sector may disrupt the supply chain in a coal-driven electric industry. This may have severe implications in the electric utility finance as, under such disruptions and fuel availability uncertainties, some entities might have to take actions such as buying electric power from other utilities, reducing generating time, and buying coal from other sources. During the 1918 pandemic, the coal supply was disrupted due to illness. Granted, the country relied far more on coal plants for electricity then, but this still indicates it is a serious challenge for coal-fired electricity markets. In New York City, it has been reported that anthracite shipments dropped about 17% from their typical levels [36].

Current pandemic plans will not be enough to sustain the coal supply chain. Although coal is being used less and less for electricity generation, it is still a crucial component in the power generation portfolio for many energy markets. Coal workers are essential personnel working in close quarters, and they are usually considered last for preventative health measures. The coal supply chain should be given a higher priority in future pandemic planning to ensure reliable power supply. In addition to coal, nuclear is another source of power generation where there are significant deficiencies. Should a number of nuclear workers become ill or unavailable due to a pandemic, running nuclear reactors successfully may be a challenge. Hundreds of steps are necessary for nuclear refueling, and the workers who perform these tasks are highly specialized, the absence of whom may significantly endanger the electric industry [36]. Many organizations in the electric sector are highly dependent on contracted employees and vendors for essential support functions. Breach of contract situations may occur during a pandemic, which would further impact the affected entities. Cross-sector interdependencies will also affect the electric sector in the event of a pandemic.

At the national level, variations in electricity demand may have severe economic effects. If power generation exceeds the demand for electricity, electric utilities need to pay customers to increase further the electricity consumption and to avoid system imbalance and failures. This means that electricity producers may have to pay consumers to purchase their electricity. Negative electricity prices are a result of the lack of flexibility of the conventional generation system [39]. However, the price would be also affected by fuel price fluctuations due to the considerable reductions experienced during the COVID-19 pandemic and major producers' disagreements.

2) PANDEMICS AND RENEWABLE ENERGY IN THE U.S

Demand for renewable energy resources under the pandemic has been increasing. While solar and wind installations have dropped around the world, the percentage of wind and solar utilization in the power generation portfolio is at an all time high. Renewable energy production and utilization is not dependent on the current pandemic, but only dependent on the weather variations. This means that generation sites are able to throttle down their nuclear and coal generation when the sun is shining. In fact, according to the U.S. Government's Energy Information Administration (EIA), renewable consumption in the U.S. has passed coal for the first time in 130 years. During the COVID-19 lock-down, the electricity demand dropped greatly allowing renewable percentages to increase. In the US, renewable demand increased by 40% [38], [40] (see Figure 8).

More renewable energy in the generation mix reflects higher volatility, and it is harder to forecast power generation into the future. Most power in response to demand for electricity is purchased the day before or more, so having the wind be lighter than expected or having more clouds in the sky forces electric utilities to purchase power immediately. This power is far more expensive than power purchased in advance, thereby hurting the electric utilities financially. An alternative strategy is throttling renewable energy and keeping nuclear and coal constant, but at the expense of environmental challenges and higher operating costs.

The IEA forecasts that the net renewable energy additions would decline by 13% in 2020 compared to 2019 [41]. According to the Wood Mackenzie, an energy analysis firm, solar and wind installations are expected to decline 20% and 6%, respectively, this year [42]). It is expected that these



FIGURE 8. Renewable energy demand following COVID-19 lock-downs [38].

numbers will rise again as the pandemic subsides and governments try to make up for the lost time. A massive influx of capital into renewable energy is expected in the coming years. Low operating cost and incentives from all sides for green energy is likely to keep the interest in renewable roll out and implementation very high [38].

C. IMPACTS ON POWER TRANSMISSION AND DISTRIBUTION SECTORS

Pandemics harbor unique threats to the electric grid. Illness and disease alone will not cause power lines and transformers to malfunction and result in a fault scenario, but they can negatively affect the system operators and the way how the grid is operated. The primary risk involving pandemics and power grids is absenteeism for critical system operations. To deal with this issue, many electric utilities publish their own guides for pandemic response, like Ontario's planning document [14]. These documents are periodically updated with the latest science and thinking, so utilities simply have to dust off the latest version when the plan is put in place. In these documents, plans, procedures, and essential personnel are laid out. These plans also lay out plans for how work will continue with proper safety measures.

At National Grid, a utility servicing the Northeastern United States, maintenance work is continuing as normal with proper social distancing and masks. The power line crews are still continuing their work, albeit with social distancing. Engineers work from home, but they still complete field visits when necessary. Utility contractors doing more of the routine tasks have had their travel reduced, and they are doing more jobs closer to their home. This reduces the geographic spread of the virus. Utility maintenance workers, by virtue of working outdoors, is a low risk profession for virus transmission. Like many utilities, National Grid implemented strict requirements for their System operators inside the control room. Starting in March 2020, 200 workers were sequestered in a secret location. Most employees lived in the company provided recreational vehicles with access to catered meals and basketball courts. At the end of the month, a second group of 200 workers was cycled in. In between

groups, the facilities were thoroughly cleaned. This extreme living arrangement helped reduce the spread of COVID-19 [43]. In addition to utility preparations, NYISO, the independent system operator (ISO) in New York, was sequestering employees on site in response to COVID-19 threat and to reduce the spread of the virus. By keeping people in a closed bubble, there is less worry about the need for testing and social distancing. All NYISO employees would still be social distancing, however, to be as safe as possible [44]. Other ISOs had similar plans in place and have implemented such practices during the past 6 months of the COVID-19 evolution.

In addition to the utility and ISO preparations, the North American Reliability Corporation (NERC) has made preparations for reliable power transmission. NERC relaxed some of their requirements and audits for electric utilities while there is a pandemic going on. In coordination, the Federal Energy Reliability Council (FERC) announced similar relaxations. FERC also announced a plan to expedite COVID-19 related requests, and they eased some requirements for in-person meetings [45].

The other challenge for the U.S electric utilities is managing Capital Expense (CapEx) and Operation Expense (OpEx) budgets. CapEx budgets allocate money for capital improvement projects like new substations, pole replacements, and new transformers. OpEx budgets are allocated for rent, payroll, and research costs, among others. Over 2020, CapEx budgets are decreasing because there is less equipment to install and the pace of work has slowed. While this may seem positive, in the complicated world of utility budgets, this is a challenge. Electric utilities should spend as close to their respective CapEx and OpEx budgets as possible. If the budgets get out of alignment, rate filings get harder and the utility is less likely to get their requests approved. This also increases OpEx budgets in subsequent years due to aged equipment in need of repairs.

Besides the above challenges, continued overloads in residential feeders and transformers may lead to equipment aging and subsequent increased failures. Electric utilities usually have maintenance planning for routine inspection and servicing to keep components in a reasonably functional condition, to maximize their life-cycles and to avoid future failures. However, with considerable variations in load profiles, the scheduled maintenance is not optimal, and rescheduling may be required. Moreover, due to limitations in the number of field crew during a pandemic, maintenance based on regular schedules is difficult to carry out.

D. PANDEMICS AND DEMAND FOR ELECTRICITY

Power demand in the US has taken a hit by the ongoing COVID-19 pandemic. As in the rest of the world, electricity usage has shifted from industrial to residential customers. Unlike Europe, comparing load demands on an aggregate scale in the United States is notoriously difficult. U.S. electric utilities do not report their load data to a central location. On a small scale, many utilities have reported drops in



FIGURE 9. The implemented load forecasting and backcasting concepts for the ISO's electrical demand prediction during COVID-19 lock-downs.

their load demand during the COVID-19 period. According to the Wall Street Journal, Snohomish County Public Utility District reported missing their energy targets when the COVID-19 lock-downs began. The utility serves numerous Boeing Factories and just over 800,000 people. Such uncertainties have challenged the energy managers and energy traders who need to forecast the future demand [32].

1) TOOLS FOR MEASURING IMPACTS OF COVID-19 ON ELECTRICITY DEMAND: LOAD FORECASTING AND LOAD BACKCASTING

Multiple scenarios are commonly approached in the electric industry to investigate possible pathways on how to reach a desirable estimate of the electrical load consumption in the future. The *load forecasting* is the method of exploring future load that has not been yet observed or determined. The load backcasting typically refers to exploring the load realizations in the past, given the information known to date [47]. If the day-ahead temperature forecasts are used to provide the day-ahead electrical load forecasts, the results are ex-ante forecasts in the day-ahead load forecasting. If the actual temperatures of the next day are used, the results are ex-post forecasts. Therefore, ex-post forecasting is commonly considered as a subset of the backcasting [48]. Figure 9 visually describes the differences between the load forecasting and load backcasting methods. In the forecasting method, the present $P_{D,t}$ is the starting point for a future pathway until $P_{D,t+1}^{f}$. On the other hand, in the backcasting approach, $P_{D,t+1}^{b}$ describes a desirable future state. This may be a future load where there are very much lower electrical load consumption than today and/or lower electrical energy consumption levels in general [48]. The backcasting model is compelling for electric utilities and the ISO planning department in their disruptive discoveries and resource allocation practices. Using this tool, electric utilities and ISOs can envision a desired future, then plan backward to the present,



FIGURE 10. Early-stage impact of COVID-19 on the electricity demand: The case of PJM [46].



FIGURE 11. Impact of COVID-19 on the U.S. Real GDP [49].

2) EARLY- AND RECENT-STAGE IMPACTS OF COVID-19 ON DEMAND FOR ELECTRICITY: THE CASE OF PJM

Figure 10 shows the "estimated" impact of COVID-19 on the peak and energy of the daily load consumption at the PJM ISO [46]. The estimation algorithm is the solution of the long-term load forecast model for each day using actual weather conditions [46]. As the actual behind-the-meter solar production data is not available in the PJM dataset, the model assumes the average behind-the-meter solar production based on the time of the year. This provides an estimate of what the load would have been for each day if COVID-19 did not exist (P_{Dt}^{f}). In order to compute the load consumption changes (in %), the MW difference between the actual load on each day (P_{Dt}^{a}) and the estimated load under actual weather conditions (P_{Dt}^{f}) is captured [46]. The estimated impact (in %) of COVID-19 on the electricity demand is given in (2).

$$\Delta P_{Dt} = \frac{P_{Dt}^{a} - P_{Dt}^{f}}{P_{Dt}^{f}} \times 100\%$$
 (2)

VOLUME 8, 2020

As an example, the estimated load for April 2nd, 2020 from the forecast model is $P'_{Dt} = 90,873$ MW. The average behind-the-meter solar adjustment for April 2nd is 1,049 MW, and the actual load realized on the same day is $P_{Dt}^{a} = 82,867$ MW. Therefore, the estimated MW impact of COVID-19 on electricity consumption measures is -8,006 MW and the estimated percentage impact is found $\Delta P_{Dt} = \frac{-8.006}{90.873} = -8.8\%$. From Figure 10, it can be seen that during the second week of April 2020, the peak load came in on average 8-9% lower ($\sim 7,500$ MW) than what the PJM anticipated. The largest impacts on the electricity consumption in the PJM service area were observed around 10-11% on March 26th and 27th ($\sim 9, 500$ MW). On weekends, however, there are no significant changes other than slight reductions on the load consumption patterns. Weekends seem to have been impacted less ($\sim 2-4\%$). The load energy consumption was less affected, with the average weekday reduction since mid-March being 7% (\sim 140 GWH per day) compared to 8% on peak.

Moody's analytics is a trusted, leading provider of economic and financial data and forecasts for the global economy [49]. Their economic forecast reports were released on 9/9/2019, 3/27/2020, and 4/10/2020. Figure 11 depicts that the U.S. real Gross Domestic Product (GDP) has been revised downward since the vintage used in the 2020 load forecast [49]. The GDP in the first quarter of 2018 (2018Q1) was 1.0. March 2020 was the first month to significantly reflect COVID-19 impacts, and April was revised down even further [49]. Economic rebound and recovery will be dependent on the progression of COVID-19 cases as well as the medical advancements such as a vaccine. According to the Moody's Analytics forecast [49], the potential full recovery will happen by mid 2023.

The impacts of COVID-19 pandemic on load consumption and used energy patterns in the PJM footprint has regularly diminished since May 2020, as several phases of reopening were enforced [46]. The daily peak load is escalating back to that of a typical normal year. Electricity

making reasonable assumptions about emerging capacities, resources, and demand-side management practices.



FIGURE 12. Recent-stage impact of COVID-19 on the electricity demand: The case of PJM [46].

energy usage is also reviving, but more gradually. Figure 12 represents that the load was entirely down about 3.9% compared to 8.9% in April and May, which noted the significant influence of pandemic-related policies and social distancing practices [46]. The electrical energy usage drops in the August's typical daily pattern, ranged from about 1.6% to 5.3%, not including Aug. 4-6 for most of the sectors, when much of the footprint endured electricity outages related to Hurricane Isaias (see the red-colored spark in Figure 12). Comparatively, the daily electrical energy usage was down throughout May 2020 anywhere from 6.1% to 13.4% [46].

While the peak impacts had risen during the Summer 2020, they started to present reemerging signs as the impact of risen weather sensitivity fades. Figure 13 exhibits the positive correlation of the daily peak demand and daily temperature from March 1st, 2020 to September 30th, 2020. July to August weekday impacts were realized 1.9% less $(\sim 2, 600 \text{ MW})$ than that predicted. Through September 2020, Figure 13 indicates that the daily peak load increases to 3.4% (~ 3, 400 MW) though with noticeably larger impacts in the second half of the month. The energy impacts live larger than the peak load, but reveals the same trend. July to August energy peaks were 4.1%, ending to 4.9% through September.

3) EARLY- AND RECENT-STAGE IMPACTS OF COVID-19 ON DEMAND FOR ELECTRICITY: THE CASE OF CAISO

Various California counties began requesting non-essential businesses to close or limit activity, including restaurants and some commercial stores, and directed companies to have their employees work from home during the early-stage evolution of COVID-19 between March 17-19, 2020. The state of California issued a stay-at-home order to all individuals living in California except for critical infrastructure sectors and essential businesses on March 20, 2020. Since the first



full week of the state-wide stay-at-home order, the California ISO (CAISO) has observed the following changes in electrical demand on weekdays and weekends. CAISO utilizes the backcast analysis given the underlying weather conditions and the type of the day (see Figure 9). The load reductions are analyzed by comparing the actual realized load with the expected values if no COVID-related order were in place [50]. Figure 14 shows that the average weekday



FIGURE 13. Weekday peak loads vs. ambient temperature from March 1st to September 30th, 2020: The case of PJM [46].





FIGURE 15. Recent-stage impact of COVID-19 lock-downs on the electricity demand: The case of CAISO [50].



FIGURE 16. Day-ahead and fifteen-minute energy prices after and before COVID-19 lock-down: The case of CAISO [50].

electricity consumption in California has reduced by 2.4% to 5.0% during the peak hours. The hourly average expected load differences for July range from -3.5% to +2.0%, with the most considerable reductions observed from 7 PM through 12 AM. The average load also reduced from 1.1% to 1.9% during peak hours over the weekends [50]. The state of California launched its pandemic road-map on May 9, which allowed counties to submit plans for the reopening phase of businesses, schools, and public places. While the complicated load forecast models could not have predicted the impact of stay-at-home order on the electrical load profile, the CAISO proceeded to fine-tune its models to improve the forecast accuracy in day-ahead and real-time markets as COVID-19 conditions emerged. The CAISO continued to observe that during days with warmer temperatures, less load reductions were observed in the evening ramp and peak hours (Actual), compared to the scenario if COVID-19 did not exist (Expected). Figure 15 shows that during the last week of June and the first week in July 2020, CAISO recognized a very partial load reduction (Actual) compared to the what was expected to be realized if COVID-19 was not persisting [50].

Figure 16 indicates that the CAISO's energy prices declined by about \$10/MWh in the day-ahead and real-time markets after the stay-at-home orders are enforced. With higher demand for electricity since the end of May, electricity prices have increased to pre-provisioned levels [50]. CAISO has no seen major impacts on transmission and distribution grid reliability from the stay-at-home orders.



FIGURE 17. Early-stage impact of COVID-19 on the electricity demand: The case of ERCOT [51].



FIGURE 18. Forecasting error of COVID-19 early-stage impacts on the electricity demand: The case of ERCOT [51].



FIGURE 19. Recent-stage impact of COVID-19 on the electricity demand: The case of ERCOT [51].

4) EARLY- AND RECENT-STAGE IMPACTS OF COVID-19 ON DEMAND FOR ELECTRICITY: THE CASE OF ERCOT

The Electric Reliability Council of Texas (ERCOT) uses a backcasting model to evaluate the load consumption, which compares the model outcomes using actual weather parameters (expected values), versus the actual hourly load realized. The difference between the actual and anticipated loads is analyzed with the error metrics. The COVID-19 impacts in April 2020 continued to lower the daily peaks, while electrical energy usage appeared to remain the same as that in the prior week [51]. Figure 17 shows that the daily peaks decreased by 4 to 5%, except on 4/24/20, the Texas hottest day in 2020. The weekly energy usage decreased by 4 to 5%, while the electrical load consumption remained consistently lower during the early morning hours between 6 AM and



FIGURE 20. Forecasting error of COVID-19 recent-stage impacts on the electricity demand: The case of ERCOT [51].

10 AM. These loads are 6 to 10% lower than the normal load that the model typically predicts if COVID-19 was not in place. The largest errors observed between the backcast outcome and the actual realized electricity consumption have been occurring at 7 AM and 8 AM, shown in Figure 18. The forecasting error is calculated in (3),

$$\varepsilon_{Dt} = P^b_{Dt} - P^a_{Dt}.$$
 (3)

ERCOT observes no pandemic-related impacts on the daily peak demand in the recent stage of COVID-19 evolution, while the weekly energy usage decreased by less than 1% (see Figure 19). Load reduction during the early morning hours between 6 AM and 10 AM appears to be shrinking. According to Figure 20, the largest error observed between the backcast and the actual load consumption was similar to the previous week in late June 2020. Most massive errors have been occurring at 7 AM and 8 AM.

V. COVID-19 AND INTERDEPENDENT NETWORKS

Comparing the pandemic response in 1918 to that in 2020 will not give an accurate representation of planning. Since 1918, networks have become incredibly interdependent, and crucial infrastructure is commingled together for efficiency [52]. Each of these sectors have their own challenges, but losing one sector could be detrimental for the other. For instance, many electric substations communicate with each other over the communication network. Even if the power grid stays online, losing the communication channels could be detrimental, resulting in potential loss of electricity to enduse consumers and mission-critical systems and services. In 1918, power grids were not as integrated as they are today. Many cities and towns were lighted through power islands, and they did not have interconnections. That is, an electricity outage in one area would have minimum meaningful impacts on the neighbouring cities. The way how the electric grid is structured has changed dramatically over the past decades. Losing a city or an electric substation would wreak havoc on the other surrounding grids. Even if the power demand was able to be met adequately, the system frequency could become unstable and the system would self destruct. Since 1918, the energy sector jobs have become more complex and specialized. This is the same across all industries, and it does not bode well for replacing critical employees in a crisis. This means that the top down response effective in natural disasters and crises is harder. The governors and Presidents who are responsible for managing emergency response may find themselves without the necessary skills to manage electric utilities and other critical infrastructure [53].

The COVID-19 epidemic affected almost every single element of the human life. With the pivotal role of the electricity grid in our modern society back-boning a number of interdependent lifeline networks, exploring the impacts on some interdependent lifeline networks can shed some light on the possible consequences in power systems (see Figure 21). In this section, the disruptions caused by the pandemic on the telecommunications, water systems, supply chains, and healthcare sectors are explored with their repercussions on electric power systems.

A. TELECOMMUNICATION NETWORK

The telecommunication (Telecom) network is a backbone connecting our society. The social distancing and stay-athome policies resulting from the COVID-19 pandemic have highlighted the importance of this network not only for the industry but also for individuals [54]. The COVID-19 impact on telecom ranges from conspiracy theories relating 5G to the spread of COVID-19 to the necessary new connectivity infrastructure in residential homes. While absurd at the first sight, the conspiracy theory linking 5G to COVID-19 in social media platforms led to the burning of 5G towers in the United Kingdom [55].

Abruptly, the COVID-19 has sent tens of millions of whitecollar workers to work remotely. Domestic and global business demand migrated from commercial to residential spaces, highly increasing the demand for mobile communications and networking infrastructure in many neighborhoods not prepared for that [56]. People working at home are part of industries, such as education and entertainment, built upon connectivity. Students and professors are at home with a clear need for high-quality internet connection to enable all types of synchronous and asynchronous online learning methods [57], [58]. With adults and kids at home, online games and streaming are another source of demand for connection [59]. However, simultaneous with this revolution, the Broadband Commission [60] has called attention to the fact that more than 3.6 billion people around the globe remain offline.

Healthcare and other industrial supply chains also heavily rely on the telecom networks to fight and adapt to the COVID-19 as it evolves [54]. Drones, cellular positioning systems, global positioning systems (GPS), radio frequency identification (RFID), and Wi-fi can be used to monitor temperature, enforce social distancing, trace infected patients, and maintain social distance [61]. The Internet of Things (IoT) is another technology which can usefully connect devices to the Internet with implications to e-commerce and its supply distribution [62]. These disruptions on telecom networks have a significant impact on power system

IEEEAccess



FIGURE 21. Electricity grid backbones a myriad of interdependent lifeline networks in modern societies.

networks. The main one is the obvious change in the electricity demand from commercial to residential sectors. The demand adjustment can be not only on the amount, but also on the time of the day that the electricity is consumed and was thoroughly discussed earlier in Section IV.

B. WATER SYSTEMS

Water system is one most energy intensive networks, an interruption in which may quickly result in a national security concern. Containment measures due to the COVID-19 pandemic have restricted people's movements, preventing them from attending schools and workplaces and traveling, and have significantly affected their normal social life and lifestyles [63]. While in some disasters and health emergencies, water can be a vehicle for bacteriological infections, this is not the case with the COVID-19 pandemic in which freshwater is a major ally to help limit the spread of the virus. This explains the exacerbated importance to have access to potable water at home — a fundamental resource to ensure adequate health conditions and to curb the spread of the pandemic – and has modified the water consumption, outflow regime, and price.

The impact of COVID-19 on water infrastructure is especially critical in low and middle-income countries, which, even before the COVID-19 crisis, faced major water access challenges. In 2017, as discussed by World Health Organization [64], close to 30% of the world population had no safely-managed drinking water services. Furthermore, 40% of people on earth presented no access to basic hand-washing facilities with soap and water available. Last but not least, 673 million people lived in areas in conditions of open defecation. Obviously, this tragedy is more concentrated in developing nations [65].

The "simple" hand-washing action, a main measure to combat COVID-19, can be just a dream for a very significant

part of the world population. Analyzing the consequences of COVID-19 in poor countries, Cooper in [66] argues that, to fix water, sanitation and hygiene (WASH) dilemmas, the design and implementation of new infrastructure policies are mandatory. Water systems are complex schemes highly dependent on electric power systems. In particular, water pumps receive a remarkable portion of the total electricity needed to run the water system infrastructure: around 4% of the total electricity usage in the U.S. is exhausted by electric pumps in water networks [67]. Investments in water infrastructure will demand support of the energy systems.

C. SUPPLY CHAIN

Supply chain (SC) response is central to battle epidemic outbreaks [68]–[70]. Cholera [71], Ebola [72], and smallpox [73] are epidemics that received sustained attention from the SC community. On March 11, 2020, the World Health Organization declared COVID-19 a pandemic [74]. Global intertwined supply networks encompass food supply, healthcare, and most service and industrial segments [59] and constitute complex production systems [75] that have been severely disrupted by the COVID-19 pandemic. Stockpiling and hoarding patterns affected the food industry during the early stages of the pandemic and resulted into a marked increase in the demand for products, such as pasta, rice, hand soap, and toilet paper [76]. Concomitantly, lock-downs and social distance created a labor shortage and logistical interruptions with repercussion on food supplies [77]. This led to a deficit of product supply in almost every segment of manufacturing industries across the world [78], which was further exacerbated by the fact that China - the world's factory - was the early epicenter of the pandemic. The reduced circulation of cars, buses, trains, and other public transportation means is another consequence of the pandemic [79], [80]. Ride-hailing

services, such as Uber and Lyft, suffered an abrupt disruption in the early stages of the pandemic [81]. Moreover, the interruption of airline flights have frozen medium and long-term distance transportation [82], [83]. Consequently, the reduction of people's mobility severely hurt supply chains.

Disruptions in global supply chains have clear and significant consequences on the demand for electricity and power system services, ranging from power balance, voltage regulation, to network maintenance [84]. NERC announced an emergency plan addressing the supply chain risks given the manufacturing slowdown in China and the preparedness to guarantee employees safety [85]. The supply problems stemming from China have negatively influenced, not only the power systems' equipment manufacturers, but also the renewable energy industry [86], [87]. We discuss next the impact of COVID-triggered SC disruptions on the healthcare and pharmaceutical industries.

D. HEALTHCARE

The COVID-19 crisis has revealed just how close to capacity most US and other countries hospitals operate, with vulnerable medical supply chains (SC) and minimal tolerance for "surge" demands [88], [89]. The health care workforce and emergency personnel are under unprecedented stress. Many front-line workers battling dual threats at work and in their home communities, while also facing economic hardship and employment insecurity [90]–[92]. The lessons of COVID-19 in systems preparedness, public outreach, and support of vulnerable communities, come at a great cost — but with the hope that they may be retained and applied to other systemic challenges in the years to come [93], [94].

The medical equipment company SC has been highly impacted by COVID-19 due to the surge in the demand for surgical masks, N95 respirators and mechanical ventilators [95]. This issue was further compounded by the prior-COVID extremely high concentration of the production of medical equipment in China, which had not enough production capacity and was struggling with the pandemic [96]. Before the pandemic, China produced 50% of the world's face masks [97]. The increasing demand and supply issues forced the U.S. to increase and accelerate the domestic production of healthcare equipment [98], to the point that the U.S. Federal Government even ordered General Motors to produce ventilators under the Defense Production Act [99].

The pharmaceutical SC is also exposed to serious concentration risks [100]. In August 2019, only 28% of the active pharmaceutical ingredients (APIs) consumed in the US were domestically produced; China and India supplied 31% of APIs for the U.S. market [101]. The significant health security risk led to increased pressure to augment the US production of APIs and other medications [102]. Clearly, this production internalization trend in medical equipment and supplies will affect the power system infrastructure in similar ways as already discussed in Subsections V-A, V-B and V-C.

VI. VISIONS FOR A PANDEMIC-PREPARED FUTURE

With the sudden shifts in electric power generation portfolios and demand profiles during COVID-19 pandemic, sustaining quality electricity supply to end customers is urgently needed. The planning, operation and control of electric power grids are impacted by "high absenteeism" pandemics in different sectors and on multiple scales. From health emergencies that could rapidly and severely limit the available workforce in electric industry for an extended and mostly unforeseeable time periods, to supply-chain disruptions leading to postponed installations and delayed maintenance implementations, the electric industry has to take all necessary actions to keep the lights on throughput the future pandemic emergencies.

The main risk factors for pandemics are spark risk and spread risk. Spark risk is concerned with how likely a pandemic is to occur in a particular area, and spread risk relates to how likely it is to spread among the populace [103]. For preparedness, it is necessary to reduce both of these risk factors. From a power grid perspective, electric utility companies were well prepared for this pandemic. By and large, there have not been any disasters, critical infrastructure failures, or other catastrophes. Thanks to previous pandemics popping up every few years, the utility companies have been well trained, and their plans were up to date. Of course, more could always be done such as drills for working from home for a week. This allows electric utilities to see what happens with their workers and verify if they have the essential tools to complete their work. In the future as more companies telework on some days of the week, this will be achieved. Cost wise, electric utilities should prioritize pandemic preparedness in the future. This means investing in keeping up to date the pandemic preparedness plans, partnering with health officials to keep prepared, and encouraging their employees to stay home if they feel sick. One main challenge with teleworking is the high dependency of the work on secure and reliable communication networks. Cybercriminals may take advantage of the crisis when more employees work remotely. Cybersecurity firms have reported a dramatic increase in spear-phishing attacks [104].

Different from the business-as-usual practices, sudden shifts in the use of electricity by the industrial, commercial, and residential sectors during the pandemics may bring about additional complexities in maintaining a continuous and reliable supply of electricity to in-need customers. Recent statistics reveal that the industrial and commercial loads have declined due to the business closures during the COVID-19 outbreak, while the residential load patterns have been trending higher as the shelter-in-place orders are in place: a slower than normal usage of electricity in the morning and increased energy use in the afternoon. Such differentthan-usual changes in demand patterns may result in power distribution transformers loading capacities being exceeded. Several challenges exist in the current practices:

• Power distribution transformers are generally aged, being in service for more than 20 years, with de-rated



FIGURE 22. The ROADMAP platform: clustering outcome for the transformer-smart meter connectivity amid pandemics.

capacities, resulting in operation closer to the limits with less energy efficiency. Existing transformers are not ready to be continuously over-stressed by the ramping in the residential electricity demand due to the evolving pandemic-shaped lifestyles.

- Transformer-customer configuration in power distribution networks was designed and planned years ago; the rising population over years and addition of new households as well as frequent feeder maneuvers and switching in power distribution network configuration have made the transformer-customer connectivity complicated or in many cases unknown.
- The existing practice does not allow recognition of "which particular household" (single- and three-phase) is being connected to "which particular transformer" (single- and three-phase connections) in the network; rather a residential "area" by a group of multiple transformers. This restricts the opportunity to control the demand as needed and limits the operational flexibility for effective transformer load management practices amid pandemics.
- The existing practices fail to holistically integrate and coordinate the objectives of feeder-transformer load management with residential household occupants' behaviors and expected thermal quality expectations. The latter encompasses the conditions inside a building

— electrical appliances, lighting, thermal conditions, ergonomics — and the impacts on the household residents' comfort.

IEEEAccess

A robust, replicable, and scalable mechanism for real-time transformer-feeder load management during pandemics is, hence, essential; a tool that can (i) estimate the real-time residential electricity demand based on the household occupancy predictions, (ii) identify which transformer (among a network of available transformers in a distribution feeder) feeds each individual household customer, and (iii) automatically perform the transformer load management practices as needed.

With the sudden shifts in the use of electricity experienced during the COVID-19 pandemic and the associated challenges it embarked on various grid operation and control practices, we suggest designing the "Real-Time Operational Awareness for Electricity Demand Management Amid Pandemics (ROADMAP)". ROADMAP is an analytical tool that can be achieved by (i) creating a Household Occupancy and Electricity Consumption (HOEC) Dataset through fieldcollected data during the outbreak, (ii) developing analytics for electricity demand estimation via residential household occupancy prediction and (iii) envisioning tools for real-time transformer-feeder load management during pandemic emergencies (see Figure 22). First, a machine learning mechanism – e.g., a hybrid of convolutional neural network (CNN) and long short term memory (LSTM) – can be proposed for

residential household occupancy prediction (RHOP) over a geographical area that utilizes the HOEC dataset and will allow an accurate estimation of the residential electricity demand. Second, a deep learning artificial intelligence tool can be applied for transformer-household connectivity identification (THCI); this is achieved through a comprehensive analyses of the Variational Auto Encoder and K-means Clustering methods using both the measured node voltages at the location of households and building footprint GIS information to characterize the sub-network cluster of households to their parent transformers [105], [106]. Third, when the first two modules are coupled, the real-time RHOP and THCI can initiate proactive decision making for online reconfiguration of the network so as to ensure an effective transformer-feeder load management, continuous supply-demand balance, availability of grid-support volt-var control functions, and required resource allocation.

VII. CONCLUSION

In conclusion, pandemic planning is incredibly important to smooth and reliable electric power delivery. As shown through previous pandemics and the COVID-19 response around the world, the necessary ingredients for a successful response are incredibly varied. There are some key takeaways which are listed as follows:

- Absenteeism is the main issue challenging the power grid operation and control. This should be addressed in future research, especially looking at how many employees an electric utility can lose before critical functions are not able to be completed.
- Pandemic planning and preparedness documents such as those used in Canada and the U.S. are very important to successful operation of the electric utilities amid the pandemic crises.
- With many of the personnel in the electric industry not available at the workplace and with the stay-at-home orders mandating teleworking and remote connections through communication networks, a pandemic may provide an ideal environment for cyber criminals to attack the power grid. Special attention and additional investments should be made on the cyber-secured platforms in the electric industry.
- Government involvement in the pandemic response can be a very helpful factor in a successful recovery. This allows for a constant and correlated response across many interdependent industries. This is part of the reason why the European response has been observed much more effective than that in the U.S.: centralized command works in crises.

Admittedly, the full consequences of the pandemic have yet to be demonstrated. Maintenance issues may take years to surface, and budgetary challenges that have been seen universally may only pose issues in a few years. The response thus far has been measurably good, but that could change at any moment. The ability for electric utilities to be flexible is going to be crucial in the coming months and years, as the grid is going to be flexed to near its breaking point. Responsible utility management will be the only mechanism keeping the lights on. Future research should also focus on the power grid planning and operation for resilience in response to weather and cyber threats while the pandemic is in place. Multi-hazard approaches and analytics could shed lights on how to tackle the simultaneous exposure of the power grid to multiple stressors and emergencies that if not properly managed, could impeded the societal growth and result in public safety and national security concerns.

REFERENCES

- WHO. Disease Outbreaks by Year. Accessed: Apr. 11, 2020. [Online]. Available: https://www.who.int/csr/don/archive/year/en/
- [2] P. Fisk. (Mar. 2020). A History of Pandemics . . . From the Antonine Plague to Black Death, Spanish Flu and COVID-19... How Disease Has Shaped Society and Healthcare. Accessed: Apr. 11, 2020. [Online]. Available: https://www.thegeniusworks.com/2020/03/a-history-of-pandemicsfrom-the-antonine-plague-to-black-death-spanish-flu-andcovid-19-howdisease-has-shaped-society-and-healthcare/
- [3] Wikipedia. List of Epidemics. Accessed: Nov. 30, 2020. [Online]. Available: https://en.wikipedia.org/wiki/List_of_epidemics
- [4] J. M. Anderies, "Embedding built environments in social–ecological systems: resilience-based design principles," *Building Res. Inf.*, vol. 42, no. 2, pp. 130–142, Mar. 2014, doi: 10.1080/09613218.2013.857455.
- [5] North American Electric Reliability Corporation and U.S. Department of Energy, "High-impact, low-frequency event risk to the North American bulk power system," in *Proc. Jointly-Commissioned Summary Rep. North Amer. Electr. Rel. Corp. U.S. Dept. Energy's Nov. 2009 Workshop*, Jun. 2010.
- [6] M. J. Roossinck. The Mysterious Disappearance of the First SARS Virus, and Why we Need a Vaccine for the Current One But Didn't for the Other. [Online]. Available: https://theconversation.com/the-mysteriousdisappearance-of-the-first-s ars-virus-and-why-we-need-a-vaccine-forthe-current-one-but-didnt-for-the-other-137583
- [7] M. Terry. Compare: 2009 h1n1 Pandemic Versus the 2020 Coronavirus Pandemic. BioSpace. Accessed: Nov. 30, 2020. [Online]. Available: https://www.biospace.com/article/2009-h1n1-pandemic-versus-the-2020-coronavirus-pandemic/
- [8] Center for Systems Science and Engineering (CSSE) at Johns Hopkins University. *Covid-19 Dashboard*. Accessed: Nov. 30, 2020. [Online]. Available: https://coronavirus.jhu.edu/map.html
- J. Nowak. (Mar. 2020). Coronavirus Puts Utility Resource Management in Focus. Power Engineering. [Online]. Available: https://www.powereng.com/2020/03/12/coronavirus-puts-utility-resource-management-infocus/#gref
- [10] Centers for Disease Control and Prevention. (Mar. 2014). 2014-2016 ebola outbreak in West Africa. [Online]. Available: https://www.cdc.gov/ vhf/ebola/history/2014-2016-outbreak/index.html
- [11] A. Badia. Ceos Express Concerns About the Zika Virus. Miami Herald. Accessed: Nov. 30, 2020. [Online]. Available: https://www.miamiherald. com/news/business/biz-monday/article96236957.html
- [12] X. Zhou. Renewables Emerge as Winner During China's COVID-19 Lockdown. Accessed: Nov. 30, 2020. [Online]. Available: https:// ihsmarkit.com/research-analysis/renewables-emerge-as-winner-duringchinas-covid19-lockdown.html
- [13] Ontario Independent Electricity System Operator. Data directory. [Online]. Available: https://www.ieso.ca/Power-Data/Data-Directory
- [14] Emergency Preparedness Task Force. (Mar. 2020). Ontario Pandemic Influenza Plan for Continuity of Electricity Operations. Issue 5.0. [Online]. Available: https://electricity.ca/wp-content/uploads/2020/03/ Ontario_Electricity_P andemic-ContOps-Planning-guide.pdf
- [15] P. Fairley, "China's ambitious plan to build the world's biggest supergrid," *IEEE Spectr.*, Feb. 2019. Accessed: Nov. 30, 2020. [Online]. Available: https://spectrum.ieee.org/energy/the-smarter-grid/chinasambitious-plan-to-build-the-worlds-biggest-supergrid
- [16] EnerData. (Apr. 2020). State Grid Resumes Construction of 800 kV HVdc Line in China. [Online]. Available: https://www.worldenergy.org/article/8291.html

- [17] C. Perkel. Covid-19 Closures: It's as if Ottawa Has Fallen Off the Electricity Grid. National Post. Accessed: Nov. 30, 2020. [Online]. Available: https://nationalpost.com/pmn/news-pmn/canada-news-pmn/covid-19-closures -its-as-if-ottawa-has-fallen-off-the-electricity-grid
- [18] L. Varro and A. Fengquan. Post COVID-19, Further Reform is Necessary to Accelerate China's Clean Energy Future. Accessed: Nov. 30, 2020. [Online]. Available: https://www.iea.org/articles/post-covid-19-furtherreform-is-necessary- to-accelerate-china-s-clean-energy-future
- [19] D. Yergin. The New Geopolitics of Energy. Accessed: Nov. 30, 2020. [Online]. Available: https://www.wsj.com/articles/the-new-geopoliticsof-energy-11599836521?mod=searchresults&page=1&pos=1
- [20] M. M. Akrofi and S. H. Antwi, "COVID-19 energy sector responses in Africa: A review of preliminary government interventions," *Energy Res. Social Sci.*, vol. 68, Oct. 2020, Art. no. 101681.
- [21] T&DWorld. China's State Grid Commissions Large Power Transmission Order. [Online]. Available: https://www.tdworld.com/digitalinnovations/hvdc/article/21121474/chinas-state-grid-commissionslarge-power-transmission-order
- [22] P. Gordon. (Apr. 2020). India's Clean Energy Development Faces Delays Due to COVID-19 Lockdown. [Online]. Available: https://www.smartenergy.com/industry-sectors/policy-regulation/indiasdevelopment-faces-delays-due-to-covid-19-lockdown/
- [23] R. Puliti. Energy Access Takes Center Stage in Fighting COVID-19 (Coronavirus) and Powering Recovery in Africa. World Bank. Accessed: Nov. 30, 2020. [Online]. Available: https://www.worldbank.org/ en/news/opinion/2020/04/22/energy-access-critical-to-overcomingcovid-19-in-africa
- [24] The New York Times. COVID World Map: Tracking the Global Outbreak. Accessed: Nov. 30, 2020. [Online]. Available: https://www. nytimes.com/interactive/2020/world/coronavirus-maps.html?auth= login-email&login=email
- [25] Red Eléctrica de España. Península-Seguimiento de la Demanda de Energía Eléctrica. Accessed: Nov. 30, 2020. [Online]. Available: https://demanda.ree.es/visiona/peninsula/demanda/
- [26] Terna S.p.A. Trend of the Italian Electricity System's Total Demand. Accessed: Nov. 30, 2020. [Online]. Available: https://www.terna. it/en/electric-system/transparency-report/total-load
- [27] Gridwatch. G.B. National Grid Status. Accessed: Nov. 30, 2020. [Online]. Available: https://www.gridwatch.templar.co.uk/
- [28] Elia Group. Grid Data-Load and Load Forecast. Accessed: Nov. 30, 2020. [Online]. Available: https://www.elia.be/en/grid-data/load-andload-forecasts
- [29] European Network of Transmission System Operators. ENTSO-E Transparency Platform. Accessed: Nov. 30, 2020. [Online]. Available: https://transparency.entsoe.eu/
- [30] A. Bahmanyar, A. Estebsari, and D. Ernst, "The impact of different COVID-19 containment measures on electricity consumption in europe," *Energy Res. Social Sci.*, vol. 68, Oct. 2020, Art. no. 101683.
- [31] M. Vanhecke. (Mar. 2020). UPDATE | Electricity Consumption in Belgium Continues to Fall. Down 16% on Average Compared to Early March, and Even Dropping 25% at Times. Elia Group. Accessed: Nov. 30, 2020. [Online]. Available: https://www.elia.be/en/news/pressreleases/2020/03/20200327_press-release-pic-de-consommation
- [32] R. Gold. U.S. Power Use Weakening After Plunging in Italy Amid Coronavirus. The Wall Street Journal. Accessed: Nov. 30, 2020. [Online]. Available: https://www.wsj.com/articles/plunge-in-italys-electricity-usehints-at-coronavirus-risks-facing-u-s-11584532801
- [33] M. Hall. COVID-19 Weekly Round-up: Residential Systems in Italy Will get a 110% Tax Rebate and UK Consumers are Being Paid to Turn Appliances on as Coronavirus Turns the Energy World Upside Down. Accessed: Nov. 30, 2020. [Online]. Available: https://www.pv-magazine. com/2020/05/27/covid-19-weekly-round-up-residential-systems-initaly-will-get-a-110-tax-rebate-and-uk-consumers-are-being-paid-toturn-appliances-on-as-coronavirus-turns-the-energy-world-upsidedown/
- [34] IEA. (2020). *Covid-19 impact on electricity*. [Online]. Available: https://www.iea.org/reports/covid-19-impact-on-electricity
- [35] Nuclear-News. German Grid and Nuclear Plant Operators Step up Coronavirus Precautions. [Online]. Available: https://nuclear-news. net/category/2-world/europe/greece/
- [36] N. Kelley and M. Osterholm, "Pandemic influenza, electricity, and the coal supply chain: Addressing crucial preparedness gaps in the united states," Center for Infectious Disease Res. Policy, Univ. Minnesota, Minneapolis, N, USA, Tech. Rep., 2008, [Online]. Available: https:// www.cidrap.umn.edu/sites/default/files/public/downloads/cidrap_coal_ report.pdf

- [37] EIA U.S. (Oct. 2019). Annual Coal Report 2018. [Online]. Available: https://www.eia.gov/coal/annual/pdf/acr.pdf
- [38] N. Mojarro. COVID-19 is a Game-Changer for Renewable Energy. Here's Why. [Online]. Available: https://www.weforum.org/agenda/ 2020/06/covid-19-is-a-game-changer-for-renewable-energy/
- [39] P. Götz, J. Henkel, T. Lenck, and K. Lenz. (2014). Negative Electricity Prices: Causes and Effects. [Online]. Available: https://www.agoraenergiewende.de/fileadmin2/Projekte/2013/Agora_Negative_ Electricity_Prices_Web.pdf
- [40] EIA U.S. (May 2020). U.S. Renewable Energy Consumption Surpasses Coal for the First Time in Over 130 Years. [Online]. Available: https://www.eia.gov/todayinenergy/detail.php?id=43895
- [41] IEA. Renewable Energy Market Update Outlook for 2020 and 2021. Accessed: Nov. 30, 2020. [Online]. Available: https://www.iea.org/ reports/renewable-energy-market-update
- [42] M. Kanevsky. U.S. Renewable Energy Generation and the Impacts of COVID-19. Accessed: Nov. 30, 2020. [Online]. Available: https://icl. coop/u-s-renewable-energy-generation-and-the-impacts-of-covid-19/
- [43] R. Moriarty. 50 National Grid Workers in Syracuse Sequestered 'to Keep the Lights on' During Coronavirus Outbreak. Accessed: Nov. 30, 2020. [Online]. Available: https://www.syracuse.com/coronavirus/2020/04/50national-grid-workers-i n-syracuse-sequestered-to-keep-the-lights-onduring-coronavirus-outbreak.html
- [44] J. S. John. New York's Coronavirus Threat Pushes Grid Operators to Work and Live in Isolation. Accessed: Nov. 30, 2020. [Online]. Available: https://www.greentechmedia.com/articles/read/coronavirus-pushes-newyorks-grid-operators-to-work-and-live-in-isolation
- [45] B. Urban and C. McCarthy. Electric Transmission and Distribution Utilities and Ferc COVID-19 Relief. Accessed: Nov. 30, 2020. [Online]. Available: https://www.energylegalblog.com/blog/2020/07/14/ electric-transmission-a nd-distribution-utilities-and-ferc-covid-19-relief
- [46] PJM. Pandemic coordination. Accessed: Nov. 30, 2020. [Online]. Available: https://www.pjm.com/committees-and-groups/pandemiccoordination.aspx
- [47] T. Hong. (Jan. 2015). Load Forecasting Case Study. [Online]. Available: https://coefs.uncc.edu/hong/files/2015/03/WTP-2015-NARUC-HONG-Shahidehp our.pdf
- [48] L. Strupeit and P. Peck, "Developing emission scenarios to aid air pollution prevention and control," Stockholm Environ. Inst., Stockholm, Sweden, Tech. Rep., 2008, p. 103.
- [49] Moody's Analytics. Economic Data & Forecasts: Global. Accessed: Nov. 30, 2020. [Online]. Available: https://www.moodysanalytics.com/ product-list/global-economic-data-forecasts
- [50] CAISO. COVID-19 Impacts to California ISO Load & Markets: March 17–July 26, 2020, Market Analysis and Forecasting. Accessed: Nov. 30, 2020. [Online]. Available: http://www.caiso.com/Documents/COVID-19-Impacts-ISOLoadForecast-Presentation.pdf#search=covid
- [51] ERCOT. (Apr. 2020). ERCOT Releases COVID-19 Load Impact Analysis. [Online]. Available: http://www.ercot.com/news/releases/show/204400
- [52] T. A. Garrett, Pandemic Economics: The 1918 Influenza and Its Modern-Day Implications, Federal Reserve Bank St. Louis Rev., St. Louis, MO, USA, vol. 90, Mar./Apr. 2008.
- [53] M. Simmons and C. Stiles. 1918 Flu Sheds Light on How COVID-19 Could Affect Critical Infrastructure. Accessed: Nov. 30, 2020. [Online]. Available: https://www.hstoday.us/subject-matter-areas/infrastructuresecurity/1918-flu-sheds-light-on-how-covid-19-could-affect-criticalinfrastructure/
- [54] H. Ahmadi, K. Katzis, M. Z. Shakir, M. Arvaneh, and A. Gatherer. (Jul. 2020). Wireless communication and the pandemic: The story so Far. [Online]. Available: https://www.comsoc.org/publications/ctn/wirelesscommunication-and-pandemic-story-so-far
- [55] W. Ahmed, J. Vidal-Alaball, J. Downing, and F. López Seguí, "COVID-19 and the 5G conspiracy theory: Social network analysis of Twitter data," J. Med. Internet Res., vol. 22, no. 5, May 2020, Art. no. e19458.
- [56] PwC. COVID-19 and the Telecommunications Industry. Accessed: Nov. 30, 2020. [Online]. Available: https://www.pwc.com/us/en/library/covid-19/coronavirus-telecommunication-impact.html
- [57] T. Gonzalez, M. A. de la Rubia, K. P. Hincz, M. Comas-Lopez, L. Subirats, S. Fort, and G. M. Sacha, "Influence of COVID-19 confinement in students performance in higher education," 2020, arXiv:2004.09545. [Online]. Available: http://arxiv.org/abs/2004.09545
- [58] E. K. Agormedah, E. Adu Henaku, D. M. K. Ayite, and E. Apori Ansah, "Online learning in higher education during COVID-19 pandemic: A case of ghana," *J. Educ. Technol. Online Learn.*, pp. 3644–3651, Sep. 2020.

- [59] M. Nicola, Z. Alsafi, C. Sohrabi, A. Kerwan, A. Al-Jabir, C. Iosifidis, M. Agha, and R. Agha, "The socio-economic implications of the coronavirus pandemic (COVID-19): A review," *Int. J. Surg.*, vol. 78, pp. 185–193, Jun. 2020.
- [60] I. T. U. (ITU). (Apr. 2020). Special Emergency Session of the Broadband Commission Pushes for Action to Extend Internet Access and Boost Capacity to Fight COVID-19. [Online]. Available: https://www.itu.int/en/mediacentre/Pages/PR05-2020-Broadband-Commission-emergency-session-internet-COVID-19.aspx
- [61] N. Saeed, A. Bader, T. Y. Al-Naffouri, and M.-S. Alouini, "When wireless communication faces COVID-19: Combating the pandemic and saving the economy," 2020, arXiv:2005.06637. [Online]. Available: http://arxiv.org/abs/2005.06637
- [62] M. Kamal, A. Aljohani, and E. Alanazi, "IoT meets COVID-19: Status, challenges, and opportunities," 2020, arXiv:2007.12268. [Online]. Available: http://arxiv.org/abs/2007.12268
- [63] G. Balacco, V. Totaro, V. Iacobellis, A. Manni, M. Spagnoletta, and A. F. Piccinni, "Influence of COVID-19 spread on water drinking demand: The case of puglia region (Southern Italy)," *Sustainability*, vol. 12, no. 15, p. 5919, Jul. 2020.
- [64] UNICEF, World Health Organization, Progress on Household Drinking Water, Sanitation and Hygiene 2000-2017: Special Focus on Inequalities. World Health Organization, Geneva, Switzerland, 2019.
- [65] J. Stoler, W. E. Jepson, and A. Wutich, "Beyond handwashing: Water insecurity undermines COVID-19 response in developing areas," *J. Global Health*, vol. 10, no. 1, Jun. 2020.
- [66] R. Cooper. (Apr. 2020). Water Security Beyond COVID-19. [Online]. Available: https://www.itu.int/en/mediacentre/Pages/PR05-2020-Broadband-Commission-emergency-session-internet-COVID-19.aspx
- [67] R. Goldstein and W. Smith, Water & Sustainability: US Electr. Consumption for Water Supply & Treatment-the Next Half Century. Electric Power Research Institute, Washington, DC, USA, 2002, vol. 4.
- [68] T. K. Dasaklis, C. P. Pappis, and N. P. Rachaniotis, "Epidemics control and logistics operations: A review," *Int. J. Prod. Econ.*, vol. 139, no. 2, pp. 393–410, Oct. 2012.
- [69] M. S. Habib, Y. H. Lee, and M. S. Memon, "Mathematical models in humanitarian supply chain management: A systematic literature review," *Math. Problems Eng.*, vol. 2016, pp. 1–20, 2016.
- [70] A. A. Anparasan and M. A. Lejeune, "Data laboratory for supply chain response models during epidemic outbreaks," *Ann. Operations Res.*, vol. 270, nos. 1–2, pp. 53–64, Nov. 2018.
- [71] A. Anparasan and M. Lejeune, "Resource deployment and donation allocation for epidemic outbreaks," *Ann. Operations Res.*, vol. 283, nos. 1–2, pp. 9–32, Dec. 2019.
- [72] I. E. Büyüktahtakın, E. des-Bordes, and E. Y. Kıbış, "A new epidemics– logistics model: Insights into controlling the ebola virus disease in west africa," *Eur. J. Oper. Res.*, vol. 265, no. 3, pp. 1046–1063, Mar. 2018.
- africa," *Eur. J. Oper. Res.*, vol. 265, no. 3, pp. 1046–1063, Mar. 2018.
 [73] T. K. Dasaklis, N. Rachaniotis, and C. Pappis, "Emergency supply chain management for controlling a smallpox outbreak: The case for regional mass vaccination," *Int. J. Syst. Sci., Oper. Logistics*, vol. 4, no. 1, pp. 27–40, Jan. 2017.
- [74] World Health Organization. (2020). Virtual Press Conference on COVID-19-11 Mar. 2020. [Online]. Available: https://www.who.int/docs/defaultsource/coronaviruse/transcripts/who-au dio-emergencies-coronaviruspress-conference-full-and-final-11mar2020.pdf?sfvrsn=cb432bb3_2
- [75] D. Ivanov, "Viable supply chain model: Integrating agility, resilience and sustainability perspectives—Lessons from and thinking beyond the COVID-19 pandemic," Ann. Oper. Res., p. 1, May 2020, doi: 10.1007/s10479-020-03640-6.
- [76] J. E. Hobbs, "Food supply chains during the COVID-19 pandemic," *Can. J. Agricult. Econ./Revue Canadienne d'agroeconomie*, vol. 68, no. 2, pp. 171–176, 2020.
- [77] S. Singh, R. Kumar, R. Panchal, and M. K. Tiwari, "Impact of COVID-19 on logistics systems and disruptions in food supply chain," *Int. J. Prod. Res.*, pp 1–16, Jul. 2020.
- [78] M. M. Queiroz, D. Ivanov, A. Dolgui, and S. Fosso Wamba, "Impacts of epidemic outbreaks on supply chains: Mapping a research agenda amid the COVID-19 pandemic through a structured literature review," *Ann. Oper. Res.*, pp. 1–38, Jun. 2020.
 [79] J. De Vos, "The effect of COVID-19 and subsequent social distancing on
- [79] J. De Vos, "The effect of COVID-19 and subsequent social distancing on travel behavior," *Transp. Res. Interdiscipl. Perspect.*, vol. 5, May 2020, Art. no. 100121.
- [80] G. Dantas, B. Siciliano, B. B. França, C. M. da Silva, and G. Arbilla, "The impact of COVID-19 partial lockdown on the air quality of the city of Rio de Janeiro, Brazil," *Sci. Total Environ.*, vol. 729, Aug. 2020, Art. no. 139085.

- [81] P. Seetharaman, "Business models shifts: Impact of covid-19," Int. J. Inf. Manage., vol. 54, Oct. 2020, Art. no. 102173.
- [82] S. Albers and V. Rundshagen, "European airlines' strategic responses to the COVID-19 pandemic (January-May, 2020)," J. Air Transp. Manage., vol. 87, Aug. 2020, Art. no. 101863.
- [83] J. B. Sobieralski, "COVID-19 and airline employment: Insights from historical uncertainty shocks to the industry," *Transp. Res. Interdiscipl. Perspect.*, vol. 5, May 2020, Art. no. 100123.
- [84] H. Zhong, Z. Tan, Y. He, L. Xie, and C. Kang, "Implications of covid-19 for the electricity industry: A comprehensive review," *CSEE J. Power Energy Syst.*, vol. 6, no. 3, pp. 489–495, 2020.
- [85] North American Electric Reliability Corporation. (Mar. 2020). NERC Takes Steps to Prevent Coronavirus Impacts. [Online]. Available: https://www.nerc.com/news/Headlines%20DL/Coronavirus%20Impacts %2011MAR20_final.pdf
- [86] K. Embree and M. Davis. Supply Chain Risks Threaten 2020 Renewables Projects. Accessed: Nov. 30, 2020. [Online]. Available: https://www.tdworld.com/renewables/article/21129615/supply-chainrisks-threaten-2020-renewables-projects
- [87] Power Technology. (Jul. 2020). Equipment Manufacturers the Most Affected in Power Value Chain: COVID-19 Poll. [Online]. Available: https://www.power-technology.com/news/covid-19-impact-highest-onequipment-manufacturers-power-value-chain/
- [88] L. Carenzo, E. Costantini, M. Greco, F. L. Barra, V. Rendiniello, M. Mainetti, R. Bui, A. Zanella, G. Grasselli, M. Lagioia, A. Protti, and M. Cecconi, "Hospital surge capacity in a tertiary emergency referral centre during the COVID-19 outbreak in Italy," *Anaesthesia*, vol. 75, no. 7, pp. 928–934, Jul. 2020.
- [89] S. M. Moghadas, A. Shoukat, M. C. Fitzpatrick, C. R. Wells, P. Sah, A. Pandey, J. D. Sachs, Z. Wang, L. A. Meyers, B. H. Singer, and A. P. Galvani, "Projecting hospital utilization during the COVID-19 outbreaks in the united states," *Proc. Nat. Acad. Sci. USA*, vol. 117, no. 16, pp. 9122–9126, Apr. 2020.
- [90] M. P. Breazzano *et al.*, "Resident physician exposure to novel coronavirus (2019-nCoV, SARS-CoV-2) within New York City during exponential phase of COVID-19 pandemic: Report of the New York City residency program directors COVID-19 research group," *medRxiv*, *Preprint Server Health Sci.*, Apr. 2020, doi: 10.1101/2020.04.23.20074310.
- [91] T. G. Kannampallil, C. W. Goss, B. A. Evanoff, J. R. Strickland, R. P. McAlister, and J. Duncan, "Exposure to COVID-19 patients increases physician trainee stress and burnout," *PLoS ONE*, vol. 15, no. 8, 2020, Art. no. e0237301.
- [92] Z. Liu et al., "Mental health status of healthcare workers in China for COVID-19 epidemic," Ann. Global Health, vol. 86, no. 1, p. 128, Oct. 2020, doi: 10.5334/aogh.3005.
- [93] E. E. Wiemers, S. Abrahams, M. AlFakhri, V. J. Hotz, R. F. Schoeni, and J. A. Seltzer, "Disparities in vulnerability to severe complications from covid-19 in the united states," *medRxiv*, to be published.
- [94] D. M. Gray, A. Anyane-Yeboa, S. Balzora, R. B. Issaka, and F. P. May, "COVID-19 and the other pandemic: Populations made vulnerable by systemic inequity," *Nature Rev. Gastroenterol. Hepatol.*, vol. 17, no. 9, pp. 520–522, Sep. 2020.
- [95] G. Gereffi, "What does the COVID-19 pandemic teach us about global value chains? The case of medical supplies," J. Int. Bus. Policy, vol. 3, no. 3, pp. 287–301, Sep. 2020.
- [96] M. L. Ranney, V. Griffeth, and A. K. Jha, "Critical supply shortages The need for ventilators and personal protective equipment during the covid-19 pandemic," *New England J. Med.*, vol. 382, no. 18, p. e41, Apr. 2020.
- [97] K. Bradsher and L. Alderman. (Mar. 2020). The World Needs Masks. China Makes Them—But Has Been Hoarding Them. [Online]. Available: https://www.nytimes.com/2020/03/13/business/masks-chinacoronavirus.html
- [98] M. S. Sinha, F. T. Bourgeois, and P. K. Sorger, "Personal protective equipment for COVID-19: Distributed fabrication and additive manufacturing," *Amer. J. Public Health*, vol. 110, no. 8, pp. 1162–1164, Aug. 2020, doi: 10.2105/AJPH.2020.305753.
- [99] M. Wayland and C. Wilkie. (Mar. 2020). Trump Orders General Motors to Make Ventilators Under Defense Production Act. CNBC. [Online]. Available: https://www.cnbc.com/2020/03/27/trump-orders-generalmotors-to-make-ven tilators-under-defense-production-act.html
- [100] P. J. Guerin, S. Singh-Phulgenda, and N. Strub-Wourgaft, "The consequence of COVID-19 on the global supply of medical products: Why indian generics matter for the world?" *FResearch*, vol. 9, p. 225, Apr. 2020.

- [101] J. Woodcock. (2019). Safeguarding Pharmaceutical Supply Chains in a Global Economy. Food and Drug Administration. [Online]. Available: https://www.fda.gov/news-events/congressional-testimony/ safeguarding-ph armaceutical-supply-chains-global-economy-10302019
- [102] E. K. Choo and S. V. Rajkumar, "Medication shortages during the COVID-19 crisis," *Mayo Clinic Proc.*, vol. 95, no. 6, pp. 1112–1115, Jun. 2020.
- [103] N. Madhav et al., "Pandemics: Risks, impacts, and mitigation," in Disease Control Priorities: Improving Health and Reducing Poverty, D. T. Jamison et al., Eds., 3rd ed. Washington, DC, USA: The International Bank for Reconstruction and Development/The World Bank, Nov. 2017, ch. 17, doi: 10.1596/978-1-4648-0527-1_ch17.
- [104] Security Magazine. (Apr. 2020). Coronavirus-Related Spear Phishing Attacks See 667% Increase in Mar. 2020. [Online]. Available: https://www.securitymagazine.com/articles/92157-coronavirus-relatedspe ar-phishing-attacks-see-667-increase-in-march-2020
- [105] R. Mohammadiziazi and M. M. Bilec, "Application of machine learning for predicting building energy use at different temporal and spatial resolution under climate change in USA," *Buildings*, vol. 10, no. 8, p. 139, Aug. 2020.
- [106] R. Mohammadiziazi and M. M. Bilec, "Developing a framework for urban building life cycle energy map with a focus on rapid visual inspection and image processing," *Proceedia CIRP*, vol. 80, pp. 464–469, 2019.



BENJAMIN WORMUTH (Student Member, IEEE) is currently pursuing the B.Sc. and M.S. degrees in electrical engineering with a concentration in energy with The George Washington University, Washington, DC, USA. In 2020, he was a Substation Engineering and Design Intern at National Grid, Syracuse, NY, USA. His research interests include wildfire resiliency, deep machine learning applications in power systems, and smart grid implementations.



SHIYUAN WANG (Member, IEEE) received the B.Eng. degree in mechanical engineering from the University of Science and Technology Beijing, China, in 2012, and the M.Sc. and Ph.D. degrees in electrical engineering from The George Washington University, Washington, DC, USA, in 2014 and 2020, respectively. His research interests include power system reliability and resiliency, smart grid and renewable energy, power grid harmonic analysis, and application of signal processing in power

systems and energy analytics.



PAYMAN DEHCHANIAN (Senior Member, IEEE) received the B.Sc. degree from the University of Tehran, Tehran, Iran, in 2009, the M.Sc. degree from the Sharif University of Technology, Tehran, in 2011, and the Ph.D. degree from Texas A&M University, College Station, TX, USA, in 2017, all in electrical engineering. He is an Assistant Professor with the Department of Electrical and Computer Engineering, The George Washington University, Washington, DC, USA.

His research interests include power system protection and control, power system reliability and resiliency, asset management, and smart electricity grid applications.

Dr. Dehghanian was a recipient of the 2013 IEEE Iran Section Best M.Sc. Thesis Award in Electrical Engineering, the 2014 and 2015 IEEE Region 5 Outstanding Professional Achievement Awards, and the 2015 IEEE-HKN Outstanding Young Professional Award.



MASOUD BARATI (Senior Member, IEEE) received the Ph.D. degree in electrical engineering from the Illinois Institute of Technology, Chicago, IL, USA, in 2013. He is an Assistant Professor with the Electrical and Computer Engineering Department, and Industrial Engineering by courtesy at University of Pittsburgh, Pittsburgh, PA, USA. His research interests include artificial intelligence and deep learning, planning, advanced optimization problems, distributionally

robust optimization, microeconomics, mathematical modeling, and multiple infrastructure assessment.



ABOUZAR ESTEBSARI (Member, IEEE) received the Ph.D. degree in electrical engineering from the Politecnico di Torino, Italy, in 2015. He pursued his research as a Postdoctoral Research Fellow with Politecnico di Torino. In 2016, he was with the E.ON. Energy Research Center, RWTH-Aachen University, Germany, as a Visiting Scholar. In August to October 2019, he was a Research Scholar with the Department of Electrical and Computer Engineering, The George

Washington University, Washington, DC, USA. From 2017 to 2020, he was an Assistant Professor with the Politecnico di Torino. He is currently an Assistant Professor with the School of the Built Environment and Architecture, London South Bank University, London, U.K. His main research interests include the power system security of supply and resiliency, smart grid applications, and multi-vector energy systems. He was a recipient of the Manby Prize ICT/GIS/BIM 2020 of ICE, U.K.





TIAGO PASCOAL FILOMENA received the B.Sc. and M.Sc. degrees in engineering from the Federal University of Rio Grande do Sul and the Ph.D. degree in engineering (major in operations research and financial engineering) from The George Washington University, Washington, DC. He is an Associate Professor of Operations Research and Finance with the Federal University of Rio Grande do Sul, Brazil. His research interest includes operations research oriented to finance.

MOHAMMAD HEIDARI KAPOURCHALI (Member, IEEE) received the Ph.D. degree in electrical engineering and computer science from Wichita State University, Wichita, KS, USA, in 2018, and the M.Sc. degree in electrical engineering from Azad University, South Tehran Branch, Tehran, Iran, in 2012. He is currently an Assistant Professor with the Department of Electrical Engineering, University of Alaska Anchorage, Anchorage, AK, USA. His research interests

include reliability, resiliency, and data analytics for interdependent lifeline systems and services.



MIGUEL A. LEJEUNE is a Professor of Decision Sciences and of Electrical and Computer Engineering with The George Washington University, Washington, DC, USA. His research interests include stochastic programming, distributionally robust optimization, decision-dependent uncertainty, energy, and financial applications.

Dr. Lejeune was a recipient of the 2019 Koopman Award of the INFORMS Society,

a CAREER/Young Investigator Research Grant from the Army Research Office, the IBM Smarter Planet Faculty Innovation Award, and the 2020 School of Business Dean's Best Senior Faculty Research Award at The George Washington University.